Author's response to reviews

Title: Best strategies to implement clinical pathways in an emergency department setting: Study protocol for a cluster randomized controlled trial

Authors:

Mona Jabbour (jabbour@cheo.on.ca)
Janet Curran (jacurran@dal.ca)
Shannon D Scott (ss14@ualberta.ca)
Astrid Guttman (astrid.guttman@ices.on.ca)
Thomas Rotter (thomas.rotter@usask.ca)
Francine M Ducharme (francine.m.ducharme@umontreal.ca)
M. Diane Lougheed (mdl@queensu.ca)
M. Louise McNaughton-Filion (lmcnoughton@mycybernet.net)
Amanda Newton (an6@ualberta.ca)
Mark Shafir (preyde.shafir@sympatico.ca)
Alison Paprica (alison.paprica@ontario.ca)
Terry Klassen (tklassen@mich.ca)
Monica Taljaard (mtaljaard@ohri.ca)
Jeremy Grimshaw (jgrimshaw@ohri.ca)
David W Johnson (david.johnson@calgaryhealthregion.ca)

Version: 2 Date: 4 April 2013

Author's response to reviews:

The manuscript has been revised to include the components in one document as requested. These include the title page, abstract, abbreviations used, competing interests, authors' contributions, author's information, funding acknowledgements, reference list, and tables. These were previously submitted in separate documents. All other requested documentation has also been included.